Author's response to reviews

Title: Genotype distribution of human papillomavirus (HPV) in histological sections of cervical intraepithelial neoplasia and invasive cervical carcinoma in Madrid, Spain.

Authors:

Benjamin Garcia-Espinosa (benjamin.garcia@mecd.es)
Ernesto Moro-Rodriguez (jemoro@uninet.edu)
Emilio Alvarez-Fernandez (ealvarez.hgugm@salud.madrid.org)

Version: 4 Date: 16 October 2012

Author's response to reviews: see over
Dear Sir:

Please find hereby attached for your consideration the review of our work entitled “Genotype distribution of human papillomavirus (HPV) in histological sections of cervical intraepithelial neoplasia and invasive cervical carcinoma in Madrid, Spain.”

Following your editorial Requests:

ER-1 Name of the Ethics Committee -Please include the full name of the ethics committee that approved the study in the 'Methods' section of your manuscript.

AA-1 Done.

The work now consist of 14 pages and 4 tables (abstract included).

Our current address, telephone and fax number are:

Area of Anatomic Pathology
Department of Histology and Anatomic Pathology
Avda. Atenas s/n
E-28922 Alcorcón (Madrid)- Spain
e-mail: joseernesto.moro@urjc.es
Tel. +34 91 488 88 65
Fax +34 91 488 89 55

We thank you for your attention and remain expectant for your response.

Yours faithfully,

Ernesto Moro Rodríguez, MD, PhD